Author's response to reviews

Title: Eating habits of a population undergoing a rapid dietary transition: portion sizes of traditional and non-traditional foods and beverages consumed by Inuit adults in Nunavut, Canada.

Authors:

Tony Sheehy (tonysheehy1@eircom.net)
Cindy Roache (cindyroache@hotmail.com)
Sangita Sharma (gita.sharma@ualberta.ca)

Version: 3 Date: 18 March 2013

Author's response to reviews:

The Editor-in-Chief,

Nutrition Journal

March 18, 2013

Dear Editor,

Please accept our revised manuscript titled, “Eating habits of a population undergoing a rapid dietary transition: portion sizes of traditional and non-traditional foods and beverages consumed by Inuit adults in Nunavut, Canada.” for your consideration for publication in Nutrition Journal. We neglected to include a title page in our previous re-submission on March 15th.

The material is original research, has not been previously published or submitted for consideration elsewhere and if accepted it will not be published elsewhere in the same form without written consent of your journal. The current study assesses food portion sizes among Inuit adults in Nunavut, Arctic Canada, and provides evidence of the nutrition transition that is occurring among Inuit in the Canadian Arctic.

The correspondent regarding this manuscript is Prof. Sangita Sharma, University of Alberta, Department of Medicine, #5-10 University Terrace, 8303 112 Street, Edmonton, AB, T6G 2T4, Canada; Phone: (780) 492-3214; Fax: (780) 492-3018; Email: gita.sharma@ualberta.ca

The authors declare that there are no potential conflicts of interest.

Thank you very much for your time and consideration.

Sincerely,

Sangita Sharma, Ph.D.
Endowed Chair in Aboriginal Health, Professor of Aboriginal and Global Health Research
Department of Medicine, University of Alberta
#5-10 University Terrace, 8303 112 Street
Edmonton, Alberta T6G 2T4
Phone: (780) 492-3214
Fax: (780) 492-3018
Email: gita.sharma@ualberta.ca